Author’s response to reviews

Title: RISK FACTORS FOR MULTIDRUG-RESISTANT PATHOGENS IN BRONCHIECTASIS EXACERBATIONS

Authors:

Rosario Menéndez (rosmenend@gmail.com)
Raúl Méndez (rmendezalcoy@gmail.com)
Eva Polverino (EPOLVERI@clinic.cat)
Edmundo Rosales-Mayor (edmundo.rosales@gmail.com)
Isabel Amara-Elori (milisaelori@gmail.com)
Soledad Reyes (reyes@comv.es)
José Miguel Sahuquillo-Arce (wadjur@hotmail.com)
Laia Fernández-Barat (lfernadal@clinic.cat)
Victoria Alcaraz (victoriaalcarazserrano@gmail.com)
Antoni Torres (ATORRES@clinic.cat)

Version: 2 Date: 27 Jul 2017

Answers to reviewers

Comments to author

Abstract: the authors are still using the name Enterobacteria, yet in their responses they stated that it had been changed.

Answers to reviewers:

This mistake has been already corrected and enterobacteria have been changed by the correct name in the new manuscript version.
Comments to author

Under results, under the subheading, follow up and outcome, the first sentence is still difficult to understand as it is not clear as to where does the number 73 come from, as it is not reflected in the Tables or Figures. If one were to make an assumption that the authors are referring to Figure 1, then one would expect the sentence to read as ’Patients who required admission were more likely to grow MDR organisms than those who did not require admission (27/153 vs 5/80, p=????).

Answers to reviewers

As suggested by the reviewer, we have changed this sentence for a better understanding in the new version.

Comments to author

The word new exacerbations still confusing to me, it would be better may be they use the word acute. Otherwise I am happy with the other responses

Answers to reviewers

As suggested by the reviewer, we have changed this word for a better understanding.

We hope to have fulfilled with all the kind suggestions made by the reviewers.

Best wishes.